---
pagetitle: hreport
title: "R `hreport` Package"
author: "Frank Harrell"
date: last-modified
format:
  html:
    monobackgroundcolor: tan
    backgroundcolor: cornsilk
    self-contained: true
    toc: true
    toc-location: left
    toc-title: Contents	
---

# hreport Package

   * [Presentation, background, and graphical elements](../../talks/rmedicine.html)
   * [Example report](report.html) using real RCT data with longitudinal data but no geographical regions
   * [Example report](test.html) using a simulated multinational trial with region/country/site
   * [Example report](testmult.html) using two simulated trials with side-by-side comparisons using tabs
   * [Video](hreport1.mkv) showing how to navigate reports
   * [Video](hreport2.mkv) showing how to understand specific graphics produced by `hreport`
   * [Video](hreport3.mkv) showing how to navigate multi-study reports
   * [Interactive combination plots](../Hmisc/combplotp.html) for showing combinations of outcomes or reasons for participant exclusions
   * [Current source package](hreport_current.tar.gz)
	 * [Latest binary packages for Linux, Windows, and Mac arm64](../bin/howto.html)
   * [GitHub repository](https://github.com/harrelfe/hreport)
   * [Change log](https://github.com/harrelfe/hreport/commits/master)
   * [greport](https://cran.r-project.org/web/packages/greport) package (previous model for producing static `pdf` reports) and [examples](../greport)

# Related Material

   * [R packages dedicated to clinical trial reporting](https://rviews.rstudio.com/2021/07/26/june-2021-top-40-new-cran-packages) - start with the line containing `clinDataReview`
   * [Automating reproducible, collaborative clinial trial document generation with the listdown package](https://journal.r-project.org/archive/2021/RJ-2021-051/index.html) by M Kane, X Jiang, S Urbanek
   * [Towards a grammar for processing clinical trial data](https://journal.r-project.org/archive/2021/RJ-2021-052/index.html) by MJ Kane
   * [Applying human factors research to statistical graphics](https://www.youtube.com/watch?v=fSgEeI2Xpdc) by John Rauser
   * [Principles of Graph Construction](/doc/graphscourse.pdf)
   * [Best Practices in Graphical Data Presentation](https://www.libraryassessment.org/wp-content/uploads/bm~doc/workshop_lyons_ray.pdf) by Ray Lyons
   * [Statistical graphics](https://hbiostat.org/doc/bbr.pdf) (BBR Chapter 4)
   * [Guiding principles for effective data visualization](https://wbuchanan.github.io/nnerppDataVisualization) by William Buchanan
   * [Evidence in support of modern statistical graphics](https://stats.stackexchange.com/questions/18431)
   * [Why scientists need to be better at data visualization](https://www.knowablemagazine.org/article/mind/2019/science-data-visualization) by Betsy Mason
   * [Safety Explorer](http://biostatdata.app.vumc.org/fh/R/other/safetyExplorer.nb.html)
   * [Statistical Tables and Plots using S and LaTeX](summary.pdf); FE Harrell (PDF with hyperlinks and bookmarks, 82 pages)

# Bug Reports
Please go to [GitHub issues](https://github.com/harrelfe/hreport/issues)
